Dear Univesity Accord,

I am attaching our submission to the University Accord Panel for consideration.

James Guthrie, Ann Sardesai and Lee Parker (2023), Submission to the University Accord Panel, 11 April 2023.

We propose to address question 5 concerning current structures and funding in higher education. Also, Q47. The Commonwealth funding for the sustainability of the Australian higher education system over the next two decades

Q5 How do the current structures of institutions, regulation and funding in higher education help or hinder Australia's ability to meet these challenges? What needs to change?

Q47 What structure of Commonwealth funding is needed for the higher education sector for the system to be sustainable over the next two decades?

Our main conclusions are that the contemporary Australian higher education system (AHES) is fundamentally flawed because public universities now treat their business model as a commercial one, not a public service delivery and crucial public good for Australian society.

We undertake this analysis by relying on Parker et al.'s (2023) working paper on the 40 years history of the Australian higher education system and several presentations by Guthrie.

At an ANU presentation recently, Guthrie (2023) pointed out that in 2021, there was much doom and gloom from university management about the prospects for the higher education sector. However, as the sector's combined financial report, Finance 2021, released by the Australian Government Department of Education in March 2023, it is apparent that talk of dark times ahead was nothing more than a smokescreen to drive down the terms and conditions of university employees and, more importantly, to make redundant over 50,000 staff.

The report makes for interesting reading. The Australian higher education system generated \$5.3 billion as a net accounting result from about \$40 billion in revenue. The report reveals that combined payments by local and international students were \$15.3 billion, whilst the government contribution for teaching and research was about \$14 billion. Also, during 2021, there was a significant increase in cash and investments held by universities as of 31 December.

That the most accrual profitable universities were Melbourne, Sydney and Monash. Nevertheless, the report suggests that the sector fared well, with total cash and investments of \$30.9 billion as of 31 December 2021, up \$6 billion or 24.1 per cent from \$25 billion in 2020.

The university VC ring-fence these assets and income from the University's operations, yet these funds have been built from operational revenue over decades. Only a tiny proportion of this \$31 billion cash and investments would be from foundations or charity gifts with special purposes attached. Specifically, ANU had cash and cash equivalents of about \$1.9 billion and borrowings \$.5 billion. This indicates a healthy financial situation for ANU.

Parker et al. (2023) provide a historical perspective of Neoliberalism and New Public Management practices in the Australian higher education system from the early 1980s to contemporary times. The Working Paper provides insights into changes in control systems and metrics applied to university teaching and research. From an audit culture framework, it examined the emerging rationales driving financially focused university governance.

In Australia, public sector universities' organisational and individual performance audits have become standard practice, couched in the conventional language of transparency, performance, and accountability. Accounting-based performance measurement, control, and audit dominate university operations, fostering mass education delivery, research outputs, and knowledge commercialisation.

Neoliberal performance management ideas have dramatically altered accounting models and internal power relations, transforming university identities and roles into a commercialised corporate model pursuing a self-interested financial bottom line aided by a private sector performance control orientation.

This transmogrification of Australian public sector university identity and role over the past forty years has had profound impacts externally on its societal role and internally on the work and roles of its academics. We argue that academics have been subject to increasingly intrusive performance measurement and control systems to which they appear to have reacted with compliance. Their internal goal displacement is reflected in their pursuit of satisfactory student teaching evaluation scores, volume and ranking of journal article publications, and research grants demanded by their employers. Performance metrics have replaced original contributions to knowledge, societal critique, and community engagement as their prime concerns.

In offering this historical case study and critique of performance management in the AHES, we pose five fundamental questions for the future.

1. Has University's commercial and reputational self-interest permanently replaced its public interest role?

- 2. Will the now predominant role of Australian university academics as production-line manufacturers of accountingised and measurable teaching and research products continue into the long term?
- 3. Have accounting academics been complicit in creating and promoting the performance control and management systems that have gradually eroded academics' and universities' broader community service role?
- 4. We contend that our historical analysis connecting past to present has revealed disturbing insights into the inner workings of the university system.

5. What of the future?

As historians, we must historiographically desist from offering predictions for the future trajectory. As Parker (1999, p.20) argued:

".....it is important to recognise that history's ability to predict is limited by the uniqueness of each event or chain of events and their surrounding circumstances which are unlikely to ever be exactly replicated at a later date. Hence history offers explanation rather than prediction, laying a basis for informing present and future decisions by way of illuminating precedents...."

Now turning to Guthrie's (2020 A, B) submissions and financial analysis to the Legislative Council Select Committee - Inquiry into the Provisions of the University of Tasmania Act 1992, he concluded that:

"In conclusion, a royal commission into the governance of the University of Tasmania will be required to achieve the broad-ranging reforms outlined in this submission. The terms of reference should focus on the lack of transparency and accountability, profit-making activities, undemocratic structures, performance management of staff, achievement of legislated purpose and performance monitoring of this, the unhealthy relationships with senior management and external consultancies and engagement with the Big Four accountancy firms."

Also, Guthrie 2020 (A, B) provided an alternative to the business accrual financial reporting and suggested a performance budget based on the functions laid down in the Act. The University of Tasmania Act 1992 specifies the University's functions, and performance against these should be reported in the annual report.

Any public discussion and accountability about the role and performance of the Council and senior management in achieving targets and expenditures per the performance budget.

6. Functions of the University

The University has the following functions:

- (a) to advance, transmit and preserve knowledge and learning;
- (b) to encourage and undertake research;
- (c) to promote and sustain teaching and research to international standards of excellence;
- (d) to encourage and provide opportunities for students and staff to develop and apply their knowledge and skills;
- (e) to provide educational and research facilities appropriate to its other functions;
- (f) to promote access to higher education having regard to principles of merit and equity;
- (fa) to foster or promote the commercialisation of any intellectual property;

Therefore we are asking for an entirely different discourse for public universities in Australia, including the University of Tasmania, which is not based on financial accrual expenses in the context of orthodox accrual accounting arising from the neoliberal corporatisation of Australian public universities over decades.

Andrew and Cahill's (2017) research highlights that alternative discourses must be developed to understand the public value of universities. The functions, as outlined in the University of Tasmania Act 1992, make clear that the University's proper functions are to be publicly accountable for "public value" activities of the University in the interests of the people of Tasmania.

The Tasmania University is a not-for-profit organisation as a registered charity with a primary function to educate citizens in a wide range of disciplines and professions and conduct research to benefit society. The Tasmania University charity commission repolt states: "During the year, the university's activities consisted of Learning and teaching, research, knowledge transfer and research training, community engagement, and activities incidental to undertaking the abovementioned educational activities."

Over the last few decades, state legislation has empowered university executives to transform the University of Tasmania into for-profit property development and investment vehicle. Enabling the reshaping of the University into an autocratic institution could be considered unethical in its treatment of staff and students.

This unethical behaviour includes the bullying and intimidation of critics, wage theft, an overpaid cadre of senior executives, and the destruction of disciplines. We understand that at this University, many well-qualified and high-achieving academics and professional staff have left or suffered burnout and the psychological trauma of being endlessly 'restructured' and surveilled by managers who appear to have no understanding or appreciation of the damage they are causing.

The relentless performance management of university staff corrodes the democratic values and principles upon which the public university system was founded and on which it should continue to be grounded. The legislated functions in the Tasmania university act clarify that public universities should serve the public interest.

However, the neoliberal policies that currently inform executive and senior managerial decision-making in Australia's public universities relegate those functions to virtual irrelevance.

The widespread adoption of accounting methodologies explicitly developed to manage the finances of for-profit corporations is just one example of how those functions have been rendered irrelevantvii. An ethically grounded accounting discourse informed by the public interest principles of transparency and accountability would restore staff and student confidence in how our universities should be run.

Distinguish Professor James Guthrie, AM Macquarie Business School, Macquarie University, Sydney, Australia, james.guthrie@mq.edu.au

Professor Lee Parker, University of Glasgow, Scotland. lee.parker.2@glasgow.ac.uk

Dr Ann Martin-Sardesai, College of Business, School of Business and Law, CQ University, Sydney, Australia, a.sardesai@cqu.edu.au

References

Andrew, J. and Cahill, 0. (2017), "Rationalising and resisting neoliberalism: The uneven geography of costs", Critical Perspectives on Accounting, Vol. 45, pp.12-28, https://doi.org/10.1016/j.cpa.2016.09.0011

Guthrie, J (2023), The Australian National University can afford to play as they have plenty of money, ANU staff address, April 2023.

Guthrie, J (2022a), Submission to the Legislative Council Select Committee - Inquiry into the Provisions of the University of Tasmania Act 1992, 25 May 2022. July.

Guthrie, J (2022b), Second Submission to the Legislative Council Select Committee - Inquiry into the Provisions of the University of Tasmania Act 1992, August.

Parker, L., Guthrie, J. and Sardesai, A. (2023), and James Guthrie (2023), performance management in the Australian higher education system – a historically informed critique, Macquarie Business School, working paper, Macquarie University, NSW 2109, Australia.

PERFORMANCE MANAGEMENT IN THE AUSTRALIAN HIGHER EDUCATION SYSTEM – A HISTORICALLY INFORMED CRITIQUE

Professor Lee Parker, University of Glasgow, Scotland. lee.parker.2@glasgow.ac.uk

Distinguish Professor James Guthrie, AM Macquarie Business School, Macquarie University, Sydney, Australia, james.guthrie@mq.edu.au

Dr Ann Martin-Sardesai, College of Business, School of Business and Law, CQ University, Sydney, Australia, a.sardesai@cqu.edu.au

Citation:

Parker, L., Guthrie, J. and Sardesai, A. (2023), and James Guthrie (2023), performance management in the Australian higher education system – a historically informed critique, Macquarie Business School, working paper, Macquarie University, NSW 2109, Australia.

Abstract

This paper aims to provide a historical perspective of Neoliberalism and New Public Management practices in the Australian higher education system from the early 1980s to contemporary times. Drawing on historical and contemporary data sets and research papers, it evaluates changes in control systems and metrics applied to university teaching and research. From an audit culture framework, it examines the emerging rationales driving financially focused university governance.

Organisational and individual performance audits have become standard practice, couched in the conventional language of transparency, performance, and accountability. Accountingbased performance measurement, control, and audit dominate university operations, fostering mass-delivery of education, research outputs, and knowledge commercialisation.

Neoliberal performance management has dramatically altered accounting models and internal power relations, transforming university identities and roles into a commercialised corporate

model pursuing a self-interested financial bottom line aided by a private sector performance control orientation.

Keywords:

Australian higher education system, neoliberalism, new public management, digitisation, performance audits, accounting, auditing and accountability, performance management systems

PERFORMANCE MANAGEMENT IN THE AUSTRALIAN HIGHER EDUCATION SYSTEM – A HISTORICALLY INFORMED CRITIQUE

1. Introduction

The idea of a self-governing and independent public university system that safeguards its academic freedoms appears to belong to a bygone era (Guthrie, Linnenluecke, Martin-Sardesai, Shen, & Smith, 2021). Instead, many of Australia's public universities have been transformed into academies of mass production for teaching local and international students (Guthrie, Evans, & Burritt, 2014; Parker, Martin-Sardesai, & Guthrie, 2021).

The model of a public university as a corporate enterprise (Shore & Wright, 2004) emerged from the neoliberal higher education policies introduced in the late 1980s. These new market mechanisms focussed on universities as income-generating enterprises that would help reduce the federal government's burden to provide higher education funding (Guthrie & Lucas, 2022). Since then, the attention of senior university management, from university principals/vice-chancellors down to an increasingly professionalised class of school deans and department heads, has been taken up with performance measurement for teaching and research, competitive quasi-market approaches to student recruitment, intense competition for research grant funding, and changing expectations over what a university should contribute to society (Martin-Sardesai, Guthrie, & Parker, 2020).

These changes are based on the accounting, auditing, and accountability practices of new public management (NPM) (Guthrie, Humphrey, & Olson, 1998). Consistent with neoliberalism, public universities have become increasingly focused on financial performance. Their organisational behaviour is now engineered through strategies, targets and elaborate measurement procedures (Parker, 2012). The techniques employed to achieve these ends—such as economic and accounting calculations and audit logic—are the same as those promoted by the Big Four accountancy firms (Shore et al., 2015). In aligning their behaviour with the Big Four, public universities have become champions of the NPM principles of efficiency, commensurability and accountability. However, critics rightly ask: Who benefits from these values? (Carnegie, Guthrie, & Martin-Sardesai, 2021; Carnegie, Martin-Sardesai, Marini & Guthrie, 2022).

Over decades, various Australian governments have systematically restructured universities according to NPM principles and practices (Olssen & Peters, 2005). Marginsons 1997). Within the Australian higher education system (AHES), the emphasis has been on the ability of universities to help create and disseminate knowledge; drive the knowledge economy by producing employable graduates; and undertake research that results in commercial innovations (Guthrie, Evans, & Burritt, 2017). Governments have shaped universities into politically and financially responsible authorities through regulatory legislation and budgetary policies. Further, neoliberal techniques for controlling universities have facilitated governance at a distance while effectively intensifying central control by the government. Consequently, most universities today have highly centralised academic management systems. These forms of accountability arguably contradict the model of universities as autonomous democratic and cultural institutions (Carnegie & Napier, 2012).

In this paper, we aim to provide a brief history of the impacts of neoliberalism on the AHES, paying particular attention to how NPM practices have influenced PMSs over the last four decades. This discussion on the changing landscape of higher education accountability practices and public policy has three objectives:

1. To identify various policy transformations in the AHES since the 1980s.

- 2. To explore the intrusion of PMS practices into Australian public university operations nationally and locally.
- 3. To reveal the implications of accountability-based performance measurement and control systems for Australian public universities on their contributions to civil society.

In pursuing these objectives, this paper offers a national historical case study of the influence of neoliberal governmental philosophies and the implementation of NPM upon university identities, roles and associated internal performance management. Accounting and its influence on performance metrics will be considered in the context of the increasingly commercialised Australian university performance management and audit cultures exhibited over recent decades.

The rest of this paper is structured as follows. Section 2 briefly outlines the methodological approach adopted in this paper, while Section 3 provides background to the contemporary AHES. Section 4 explores the literature on the neoliberal commodification of academic labour and the audit society. Section 5 contains a brief review of Australian policy since 1985, the four themes of the Dawkins reforms, the Hoare review, and how assessments of research, teaching and students have been reconfigured. Section 6 analyses the past four decades through the themes of accountability, audit, and performance measurement systems. The final section provides conclusions and policy suggestions for the recent review of developing the Australian Universities Accord in 2023.

2. Methodological Approach

In delivering this historical perspective, we have adopted an instrumentalist constructionist perspective where we reflect on contemporary events and interpret them through accounting for the past. By analysing the past, we seek relevant knowledge to inform the present (Hurst, 1981; Previts et al., 1990a). In particular, we seek to understand how past patterns of behaviour have shaped PMSs and accountability practices in the AHES (Previts et al., 1990a). The future implication of past change is also considered, as Demski (1985, p.72) argues, "Whenever we encounter a problem, it goes without saying that the problem has a past and, presumably, a future." In addressing this, we take the attitude of historiographer John Tosh (1991), who sees historical research as making visible people, practices, and outcomes. He challenges conventional wisdom and identifies past experiences and precedents that have led to the current policies and strategies. So, as argued by Previts et al. (1990b), we contend that reflections on the history of the AHES can provide valuable pointers to the ongoing trajectories of change.

Accordingly, we present a longitudinal narrative timeline account of the significant periods of change in the AHES since the 1980s, particularly identifying critical events, including the Dawkins AHES reforms, the Hoare review reforms to AHES, the Australian Research Council's reconfiguration of research assessment, and progressive changes to teaching and student assessment over the 40 years. Those significant institutional changes are then revisited via thematic analysis of the underlying agendas and strategies involving the impact of NPM, the invasion of quality audits, the intensification of performance measurement, the orientation of teaching the student as a customer, and the focus on commercial competition. This way, historical events are analytically connected to thematic reflections on current university system orientations and operations. In this way, accounts of the past are connected to reflections on current practice.

Nonetheless, we do not suggest that historical evidence will predict the future of the AHES. Instead, we believe that by contextualising the present and critiquing past events, we can identify patterns of change over time that might be useful for informing our decisions in the

present and future (Carnegie and Napier, 1996; Parker, 1999). Carr (1987) argues that we seek connections between the past and the present to gain a better sense of the trajectory of change and pursue an enhanced appreciation of the future (Parker, 1997). From a postmodernist perspective, we share Jenkin's (1991) ambition of trying to render the previously opaque more transparent so that insights from the past can contribute to some hope of emancipating the present (Parker, 1997).

In presenting this historical reflection on the past and present trajectory of AHES change and focus at the macro levels of the national university experience, the study has employed governmental reports and prior published empirical and analytical studies from the accounting and higher education research literature. The latter covers both Australian and international published research, a significant proportion having been researched and published by the authors of this paper.

In doing so, we acknowledge the exercise of reflexivity in our historical analyses and contemporary reflections as researchers with lengthy prior published research experience in this field of university commercialisation, accountingisation and performance management. As Haynes' (2017) exposition on reflexivity in accounting research points out, our analyses and arguments concerning AHES historical changes and present trajectory bear the hallmarks of our positions as longstanding researchers into university governance, strategy and control and the analyses and findings of our research have presented in the past, with respect both to Australian and international university trends. Our reflexivity acknowledgement must necessarily include accounting for our prior experience as researchers in this field and as academics in the university sector (in Australia and overseas) covering the past 40-year period (and more). This reflexivity has been exercised, mindful of our combined personal engagement in the university environment over these 40 years, occupying multiple roles from emerging to senior scholars, department headships, school deanships, research deanships, teaching program directors and more. Finally, such reflexivity also bears the imprint of our home discipline, accounting, and the lens this brings to our historical examination.

Finally, this study's approach takes up Hopper's (1999) contention that accounting research can benefit from analysing and reflecting on the broad spectrum of published international research into specific issues to build on what otherwise can become a separate set of individual studies. Accordingly, this paper analyses a considerable corpus of accumulated empirical and analytical studies, presenting a historically informed macro analysis connecting past and present.

3. The Australian higher education system

Comprising 37 public system universities and over 150 private higher education providers, the AHES is an economically and socially critical part of Australian society. On average, the AHES educates a diverse mix of 1.4 million national and international students yearly. In 2021, there was much doom and gloom from university management about the prospects for the higher education system. However, to the sector's combined financial report, Finance 2021, that talk of dark times ahead was nothing more than a smokescreen to drive down the terms and conditions of university employees and, more importantly, to make redundant over 50,000 staff.

The Australian higher education system in 2021 generated \$5.3 billion as a net accrual accounting result from revenue of about \$40 billion. The report reveals that combined payments by local and international students were \$15.3 billion, whilst the government contribution for teaching and research was about \$14 billion. Also, during 2021, there was a significant increase in cash and investments held by universities as of 31 December.

The report suggests that the sector fared well, with total cash and investments of \$30.9 billion as of 31 December 2021, up \$6 billion or 24.1 per cent from \$25 billion in 2020. The universities ring-fence these assets and income from the university's operations, yet these funds have been built from operational revenue over decades. Only a tiny proportion of this \$31 billion would be from foundations or charity gifts with special purposes attached to them.

However, this growth is primarily due to HELP loans, which domestic students repay in the future. It is estimated that outstanding student debt reached A\$74.4 billion in mid-2022, up almost A\$6 billion in a year and A\$50 billion in a decade. With inflation at 8% in 2023 and loans indexed to CPI, these numbers will be higher.

In addition, the federal government introduced the fees for students in management, commerce, law, and economics have risen by 27.7%, in creative arts by 66.1%, and in communications and humanities by an astonishing 113.1% (Hurst, 2020). Furthermore, all this has occurred because of cost claims (Guthrie, Andrews, & Baker, 2020).

The Universitas 21 Reports provide an annual snapshot of higher education systems in 50 countries. Notably, these reports assess national higher education systems rather than individual universities. In 2020 Australia was ranked ninth worldwide and eighth the year before. However, Australia was one of only two countries in the top five for output that did not rank in the top ten for resources – the other country being the United Kingdom. In this regard, Australia ranked 14th based on the resources available to its universities and alarmingly, Australia sits at 34 out of 50 countries in terms of government expenditure on tertiary education institutions as a percentage of GDP.

Table 1: Top 10 university systems and their measures for resources and output, 2020

	Overall ranking		Resource measure		Output measure	
	Rank	Score	Rank	Score	Rank	Score
United States	1	100.0	10	86.2	1	100.0
Switzerland	2	90.1	3	97.9	5	67.5
Denmark	3	85.7	4	97.6	4	67.8
Singapore	4	84.5	2	98.2	11	59.0
Sweden	5	84.3	5	94.2	6	65.1
United Kingdom	6	83.6	19	65.5	2	72.9
Canada	7	83.2	6	93.0	7	64.1
Finland	8	82.8	9	86.3	9	60.9
Australia	9	82.2	14	71.8	3	67.8
Netherlands	10	81.6	11	79.3	8	62.4

Source: Universitas 21(U21, 2022)

These impacts are due to the pervasive influence of neoliberal NPM practices and the operations model adopted by Vice-Chancellors (VCs) in AHES over the past 40 years. Under these policies, undergraduate education has become far more narrowly focused on training students for immediate short-term employment and engagement with industry rather than laying the foundations for contributing to society over the long term. At the same time, a significant proportion of university teaching is now done by low-cost casuals (Guthrie & Lucas, 2022).

In short, AHES relies heavily on international students for income in contemporary times. It employs a highly casualised workforce to control costs control and build flexibility into its labour force. Furthermore, it strategically focuses on student enrolment growth and international student fee revenues. However, even though this approach is unsustainable, it has become the default setting for political elites and university VCs (Hil, Pelizzon, & Baum, 2022; Lake et al., 2022).

4. Neoliberal commodification and audit

There are many definitions of neoliberalism, and its points of emphasis have changed over the decades. According to Brown (2015, p. 28), "Neoliberalism is most commonly understood as enacting an ensemble of economic policies in accord with its root principle of affirming free markets". Indeed, it has been argued that market fundamentalism and self-interested economic competition have become the prevailing metaphors for social and cultural relations and institutions (Brown, 2015; Scharff, 2015).

When put into practice, neoliberalism aims to reduce the size of the public system and reassert political control by introducing NPM and new public financial management (Guthrie, Humphrey, & Olson, 1998; Guthrie, Olson, & Humphrey, 1997; Guthrie, Olson, & Humphrey, 1999). In addition, it advocates an individualistic approach to accountability where citizens should be self-managing and self-enterprising rather than promoting collective responsibility (Morrish, 2017).

Neoliberal reforms have prompted various government-led public initiatives in higher education in Australia, the UK and internationally (Vodeb et al., 2022). From this foundation, governments have propelled higher education institutions towards a commodified redefinition of their identity and societal role. As a result, higher education responsibility has been detached from the public domain and re-attached to the market and private enterprise (Davies, Gottsche, & Bansel, 2006). Education is not seen as distinctive or unusual; it is treated as a product or service like any other, and its value is determined by the so-called marketplace (Parker et al., 2021). Nor have universities been exempt from the neoliberal ideal of personal responsibility. Thus, a culture of entrepreneurialism has been enforced, where surveillance mechanisms proliferate, financial accountability requirements constantly escalate, and efficiency targets abound (Parker et al., 2021). By way of an aside, while these aspects may have proliferated in the AHES, Australian press articles in the 2020-22 period have begun to raise questions about university risk management and risk disclosures regarding their exposure and responses to such dramatic losses of international student revenues as occasioned by the pandemic and associated federal and state level Australian government responses and controls (Carnegie, Guthrie and Parker, 2020; Carnegie and Parker, 2021).

Nevertheless, over time, government funding for Australian higher education has, in real terms, been reduced. Parker (2011) referred to this as a clever sleight-of-hand, where the government reduces funding while imposing more interventionist performance controls over universities. For example, he identified that AHES student enrolments had risen from 393,734 in 1987 to 957,176 in 2005, while the proportion of higher education funding from the Commonwealth government fell from 85% in 1987 to 41% in 2003. Alongside developments such as this, there have been greater demands to account for how funding is used – accounts that rely on the elaborate neoliberal paraphernalia of auditing, accrual statements, and other controls (Guthrie & Lucas, 2022). Within universities, the meaning of efficiency has shifted away from a generalised social and economic good towards a notional monetary value attached to designated products and practices. Thus, graduate students and published research have been redefined as primary products (Hil et al., 2022).

Moreover, this new university world of commodified products and services has become subject to monetised measurement and calculation. For instance, everything from the university to its teaching, research, and graduate employment is ranked and rated. Furthermore, these rankings and ratings are primarily produced and controlled by external private companies like publishing houses and ranking organisations.

A culture of time charts, benchmarks, league tables, and vision statements is embedded in the models of academic production to the extent that, according to Halffman & Radder (2015, p. 896), what is at stake is the "very notion of knowledge itself". These metrics undermine the professional autonomy of the academic and the values of independent critical inquiry. Moreover, they affect academic life and intellectual independence (By, Burnes, & Oshwick, 2013). Academic freedom is intrinsically linked to the notion of the university as a public good. It is central to higher education and affects all aspects of university work (Latif, 2014). Academic work is understood to be advanced through the "creativity ... originality [and] freedom" of academics (Henkel, 2005, pp 149-150). It is epitomised by an absence of interference and idealised in the freedom to pursue inquiry and publish findings (Finkin & Post, 2009; Liszka, 2008). Freedom to pursue one's research interests is a significant attraction for academics (Bexley, James, & Arkoudis, 2011). However, managerialism and an audit culture undermine these academic freedoms. They often erode the ability of academics to exercise their professional discretion and expertise in choosing areas of research they consider relevant, potentially significant, or worth pursuing. Shore and Wright (2015) define audit culture as the process by which accounting and financial management are employed in the governance of people and organisations and the degree to which their measurement and ranking practices have been financialised, institutionalised and become all pervasive – even in the routine language of organisational life. They point to such audit culture's social and cultural consequences as major issues.

The surveillance of academics by university audit culture and calculative practices has rightly become an object of study – and activism – mainly because of the increasing use of digital technologies to monitor and control employees. Measuring academic outputs, quantifying scholarly work, and controlling culture have become so ingrained that auditing has become a central organising principle (Shore & Wright, 2015). Performance audits of the university and its employees are not just performance management and governance techniques; they are a mode of thinking. To argue that their activities are evidence-based, ranking organisations overproduce quasi-empirical language (Shore & Wright, 2015). Rating systems and league table comparisons have become popular. The scale to which performance indicators and audits have spread in contemporary times is extraordinary, and universities and society have embraced and endorsed them (Connell, 2022). Audits appear to be natural and benign solutions to performance management and governance (Shore et al., 2015). However, Power (1997) suggests that while audits and indicators may promote organisational transparency, they become opaque. Indicators become targets as institutions are reshaped according to the criteria and methods used to measure them. Hence, universities and academics alike are transformed into auditable entities that focus their energies on doing "what counts" (Carnegie, 2023).

It is worth observing that Harvey (2005, p. 162) identifies another characteristic of neoliberal institutions as "the management and manipulation of crises". Anyone who has worked in Australian universities since 1980 will recognise that their careers have unfolded in an era of constant crisis management, accompanied by urgent calls for 'change', 'efficiency', and 'modernisation' to forestall further turmoil (Sims, 2022; Tregear et al., 2022). Morrish (2017) argues that academics have gained nothing from pursuing corporatisation and that their silent acquiescence to the process has made them collaborators. Canaan (2010, p. 58) asks, "Why do we all just keep going along with it?" suggesting that the answer is because the carrot of self-

actualisation is dangled before academics while their compliance is ensured with the stick of internal regulations. Whatever the root cause, it is undeniable that, as governments have transformed universities into neoliberal institutions, academics have been hard-pressed to generate a collective position of resistance (Hil et al., 2022).

5. Post-1980s national higher education policy transformation in Australia

Our review of Australian higher education policies since the 1980s, highlights that the AHES has not been immune to neoliberal commodification and audit practice encroachments. Financial quantification, performance management, control and accountability have been central features in the policy changes implemented, and these tenets have been embedded in a series of reforms. The changes started with the Dawkins reforms of the late 1980s and subsequent policy and funding developments, as outlined in the following sub-sections.

5.1 The Dawkins reforms

John Dawkins, the then Labor Federal Minister for Education, was focused on clarifying the federal government's responsibility for higher education institutions (Bay, 2011). Responding to the introduction of major neoliberal reforms in the 1980s under NPM, the AHES moved to quasi-market-based operations for teaching and awarding degrees (Guthrie & Neumann, 2007). The Dawkins changes then led to abolishing the then-binary higher education system (consisting of colleges of advanced education (CAEs) and universities) and establishing a unified national system for higher education. The Federal government's focus was also on developing budget management structures and a changeover by universities to an accrual audit and reporting system to emphasise efficiency (Guthrie & Parker, 1990).

In the binary system, universities were funded based on their teaching and research activity, and CAEs were funded according to their number of approved courses. However, introducing this unified national system required CAEs to amalgamate with existing public universities (Coaldrake & Stedman, 1999). The consequent demise of the binary system led to a reduction in the number of higher education institutions from 88 (19 universities and 69 colleges) to 36 universities by 1996 (HEC, 1988; Mahony, 1992, 1993).

The new system also required each university to have a standardised single governing body, one chief executive, one educational profile, one funding allocation, and one set of academic awards (Gamage, 1992; Marginson, 1997). The system was based on the assumption that it would be easier for the government to govern the system under a unified approach (Bessant, 1996). The preference was for "strong, decisive implementation ... [and] policies by institutional managers" (Dawkins, 1988, p. 101). The reforms encouraged universities to adopt smaller governing bodies comparable to the boards common to private businesses. These 'specialised management skills' were essential to university management (Dawkins, 1988).

Another objective of the Dawkins policies was to prescribe a review of institutional management to ensure systems of accountability and performance measures were developed (Dawkins, 1987). The subsequent Dawkins (1988) *White Paper* foreshadowed the changing accounting mechanisms, performance indicators, and audit practices for higher education funding arrangements and the university profiling process. In addition, research was to be funded increasingly through competitive grants with the "goal of maximising the research potential of the AHES and achieving a closer alignment with broader national objectives" (Smith, 1989, p.1). Thus, the defined role of research within the AHES in the 1990s was reorganised, and research funding was to be allocated based on competitive principles.

5.2 The Hoare Review

The Hoare Review (Hoare, 1995) was the next major reform in the AHES. This review recommended that universities adopt contemporary governance, managerial capacities, and workplace practices. Performance measurement, centralised management, and cost reductions were the core principles of managerialism translated into Australian public universities by the Hoare Review, thereby diminishing the scope for academic authority and collegial processes. In particular, the newly introduced professionalised organisational system of university governance, imported from the private system, offered a process that could break existing academic power structures. In this period, power struggles between academics and administrators emerged in response to the rise of managerialism, especially regarding finance and resource allocation (Lake et al., 2022; Ryan & Guthrie, 2009).

To cope with conflicting pressures, university VCs often opted for a new professional-managerial class to control operations and finances. Under managerialism, performance and performance management came to be seen more in quantified economic and financial terms than in qualitative social terms. Thus, during the 1990s, universities imported business accounting and management control techniques. These included establishing quasi-markets, fostering employee competition, marketising public system services, and developing efficiency measures (Parker & Guthrie, 1998).

5.3 Reconfigured research assessment

Since its establishment in 1987, the role of the Australian Research Council (ARC) has been to provide (along with other government agencies) both research funding and research policy advice to the AHES (Harman & Meek, 1988). However, the *Australian Research Council Act 2001* transformed the ARC into an independent body. As a result, the ARC granted a broader range of advisory functions and full administrative responsibility for assessing grant applications (ARC, 2012). The ARC also became responsible for various research support schemes, moving research funding mechanisms away from indirect means (e.g., through the core funding of the AHES) to direct and competitive financing of individual research projects and researchers (ARC, 2016).

Over the ARC's history, different performance measures and indicators have recorded and measured research activities (Martin-Sardesai & Guthrie, 2018). This has included the Relative Funding Model, the Research Quantum, the Institutional Grants Scheme, and other models focused on formulae-driven, project-based funding and performance-based block research grants. These changes have had significant impacts on university research and the manner of its accounting and financing. The ARC's declared priority research areas used to be determined by federal government guidelines (e.g., by Ministers). These guidelines dictated the distribution of the research quantum and the use of associated PMSs. Now the ARC announces its priority research areas and associated PMSs annually. These announcements directly impact how universities operate, with many adopting a corporate business focus on performance measures, audits and controls to secure ARC funding.

5.4 Reconfigured teaching and student assessment

Student evaluations of university teaching have a relatively recent history in Australia, notably triggered by the 1964 Martin report into Australian higher education. In 1969, there was a call by the increasingly organised National Union of Australian University Students for teaching and learning units and teaching abilities to be factored into academic tenure and promotion decisions (Darwin, 2015). Additionally, academics were required to hold compulsory teaching qualifications. These calls came in response to growing tertiary student dissatisfaction with teaching quality. Factoring teaching abilities into tenure and promotion decisions was a

particularly controversial demand that attracted attention across universities. This issue was still not resolved by the early 1980s (AVCC, 1981). However, Dawkins's (1987) changes to the AHES reorganised the university system and introduced student assessment of teaching as a staff performance metric. Hence, students evaluating teachers was effectively mandated by an accord between the Australian government and the reorganised Australian universities. This essentially transmogrified students into education consumers (Darwin, 2015).

The year 2000 saw the introduction of a quality assurance framework (Commonwealth of Australia, 2000). The framework included: external 5-year cyclical quality audits by the Australian University Quality Agency; federal government monitoring of university performance across enrolments, student experience, academic outcomes, and graduate employment; and accreditation compliance with various laws, regulations and the Australian Qualifications Framework. In 2008, the Bradley et al. (2008) review of higher education, however, found the system highly complex with an excess focus on quality assurance systems and processes rather than outcomes (Shah, Nair and Wilson, 2011; Shah and Jarzabkowski, 2013). As such, quality assurance policies for teaching and strategies driven by the government have been a continually-evolving phenomenon in the AHES landscape – a phenomenon that involves ever-expanding bureaucracies and audits to secure performance-based funding in a period of funding declines and shortfalls (Shah, Nair and Wilson, 2011; Shah, Lewis and Fitzgerald, 2011).

From the 1980s onwards, government and university management's general focus has been graduation and completion rates. Now, four decades later, universities pursue various statistics and performance metrics. To name just a few, local and international student loans, student liability status, trends and Commonwealth-supported places, enrolments in courses leading to professional registration, work-integrated learning, student satisfaction with teaching and university experience, student outcomes and employment, award course completions, attrition and completion rates, graduate salaries, jobs by industry and occupation, and completions by the skills priority list, along with many others. Performance management systems such as those that grade students' experience of lecturers and study units now observably drill down from organisation-wide metrics to the individual academic. Such metrics are often presented as scorecards, some publicly available by the universities because management thinks highly-rated academics will attract more students to their subjects.

What is observable is that government funding is not based on any of the above multiple metrics other than federally-supported places, an input activity, and capped university places, an input criterion. Today, in public universities, most local students pay a substantial part of their higher education costs through the Higher Education Contribution Scheme (HECS) loan scheme, and international students pay whatever the university can charge (this can be \$100,000 for a Master of Commerce at Sydney). Experience shows that university management highly prizes the latter source of revenue as an alternative source of funding to the still limited funding supplied by the government. Students reflecting their apparent perceptions of a university revenue dependence on their fees have been observed to often employ course evaluations as an avenue for highlighting their disappointments. These can take the form of unfair evaluations of courses and teach in reaction to performance standards required and results expected or achieved by students in the latters' attempt to attribute their performance to "poor teaching" rather than "poor learning". In a commercialised university environment where the student has been transmogrified into a fee-paying customer, students often appear to avoid accepting accountability for their efforts and performance (Lama et al., 2015). In response, staff has observed university management as increasingly caving into the critiques levelled by students,

who are frequently regarded as more reliable than academic staff. Such gaming has arguably become a significant feature of university culture and life, risking undermining the organisational culture of universities as learned institutions.

6. Accountability, audit and performance measurement systems

As seen above, the intrusive practices of accounting, audit, accountability and performance measurement in Australian public university operations has had far-reaching effects over time. In particular, four impacts are worth exploring: the imposition of managerialism, the audit explosion, national research assessment exercises, and pre-and post-COVID-19 impacts. These areas have gained prominence because there has been an erosion of trust between the Australian government and public universities and because various research and teaching assessment exercises have introduced new performance criteria and accountability rules. In response, public universities have imposed strategic monitoring of faculties, departments and individual academics. As a necessary condition of the managerial model, any resemblance of collegial democracy has been weakened or eliminated (Baum, Dollard, Fisher, Freeman, & Newman, 2022).

6.1 The onset of managerialism and new public management

The reforms of the 1980s are referred to as 'managerialism' in the public system accounting literature (Guthrie, Parker, & Shand, 1990; Parker & Guthrie, 1993). They include five characteristics (Guthrie & English, 1997; Martin-Sardesai, Guthrie, Tooley, & Chaplin, 2019):

- clear and consistent objectives that incorporate strategies, plans, performance agreements and individual programs;
- greater managerial autonomy through the delegation of ministerial authority and the devolution of managerial authority to senior management;
- performance evaluations through performance indicators at the national, organisational and individual programme levels;
- the introduction of accrual accounting statements for external accountability; and
- rewards and sanctions for senior public service managers.

This managerialist trend has gone hand-in-hand with the gradual commercialisation of Australian universities whose managements strive to comply with Australian government economic agendas and generate massive financial returns from overseas student recruitment. Thus, the core strategic values of most universities have evidenced a gradual shift towards market share, competitive positioning, financial returns, industry partnerships, vocational relevance, and customer responsiveness (Parker, 2002). Image, brand and customer service have become a language and an ethos that increasingly permeates the university lexicon while, despite public protestations to the contrary, service to the public interest has been gradually replaced by priority service to government, industry, commerce and the university's own financial and growth objectives (Parker, 2011; Carnegie, 2022).

Supported by neoliberal rationality, managerialism has become manifest in NPM practices. It has filtered into universities through the financial management improvement programme (FMIP), an umbrella mechanism to implement managerial reforms throughout the Australian federal public system (Department of Finance, 1994). Managerialism has introduced a distinctive mode of governance into universities in Australia – one that is focused on financial and operational accountability. Neoliberal techniques allow the federal government to maintain governance at a distance while university management exerts control through highly centralised management processes (Marginson, 1995). Top-down management controls and performance measurement have influenced everything from university executive management to individual academics. These practices have positioned Australian academics as self-

managing individuals, supposedly "free agents [that] are empowered to act on their behalf while 'steered from a distance' by policy norms and rules of the game" (Marginson, 1993, p. 63). They have become passive employees subject to top-down, authority-based management and intrusive individual performance controls. The multiple performance measures applied to academics were observed as early as two decades ago by Parker (2002) when he identified the neoliberal pressures for measurable individual performance outcomes. These, he argued, produced a self-focussed individualistic approach to academic work; academics began focussing their efforts on tasks that are measured and rewarded.

6.2 The audit explosion

Consistent with neoliberalism, the Dawkins Reforms of 1987, instituted by the federal Labor government, ushered in sweeping changes to Australia's higher education system that spanned management structures, pricing, the devolution of budgets, auditing mechanisms, reporting systems, and more—all in the name of improving service delivery, efficiency, and effectiveness (Guthrie & Parker, 1990). NPM emphasises accountability and efficiency through explicit quantitative performance measures and external audits, including performance reviews, staff appraisal systems, performance-related pay, quality audits, customer charters and quality standards (Parker et al., 2021).

There were three rounds of Quality Audits between 1993 and 1995, promising additional funding to incentivise institutions to conform with the government's priorities (Gallagher, 2003). From 1996 on, quality issues were incorporated into each institution's reporting obligations (Gallagher, 2003, p.30). Power (1991) phrases this regime as "the audit explosion". It is a regime that requires expert judgment about which 'realities' are to be selected for examination. It also narrows attention to only the auditable aspects of performance, excluding other aspects from view. At the same time, auditing has become a standard way of dealing with and absorbing various social problems (Power, 1997). Governments believe in transferring market properties to public services because they see NPM delivering benefits in the guise of efficiency—tightening accountability and social control. Power (1997) calls this the "audit society."

The Australian government has employed performance management systems to steer the AHES towards teaching and research performance. Inherent within these systems is a requirement for universities to be more responsive to the strategic imperatives of the government. Their inception has taken the form of explicit teaching, research performance indicators, and quality audits. Universities, in return, have been required to meet accountability demands through performance reporting and quality audits by government agencies. These expectations have been monitored through performance targets and other accountability mechanisms, such as audited accrual financial statements, enterprise profiles, and quality audits through the Australian Universities Quality Agency (AUQA) and Tertiary Education Quality and Standards Agency (TEQSA).

6.3 Research assessment exercises: an intensification of performance measurement

Coinciding with the emergence of global university ranking systems (Harman, 2009), the Australian Government in 2006 announced that it would be establishing an assessment/audit of research output in the form of the Research Quality Framework (RQF) (Martin-Sardesai, Irvine, Tooley, & Guthrie, 2015). From its inception, the RQF was pilloried as "poorly designed, administratively expensive and relying on an impact measure that is unverifiable and ill-defined" (Carr, 2008). Although a change in government in 2007 led to its cancellation, the new federal Labor government in 2008 announced a replacement for this national research assessment scheme called Excellence in Research for Australia (ERA). Enacted in 2010 across

all Australian universities, the ERA is a quality assurance process that uses metrics (or other agreed-upon quality measures) appropriate to each research discipline. The AHES has engaged in subsequent ERA exercises in 2012, 2015, and 2018 (Martin-Sardesai et al., 2019).

Early on, the ERA focused on rating research and ranking journals and publishers. National and international rankings (or league tables) from these assessment exercises now play a role in managing universities (Martin-Sardesai, Irvine, Tooley, & Guthrie, 2017c). However, what is less explicit is that implementing ERA 2010 produced a potent form of a governmental audit. Its outcomes (in the manner of ratings) have incentivised a strategic and operational response by the management of Australian public universities to enhance the quality (and quantity) of research consistent with ERA requirements. To this end, universities have instituted highly instrumental PMSs covering internal research, individual academics and disciplinary groups (Martin-Sardesai et al., 2017c).

In pursuing higher ERA scores for various disciplines, several universities have manipulated these measurements, increasingly quantifying research outputs at the individual and departmental levels and embedding those metrics into their PMSs (Martin-Sardesai, Irvine, Tooley, & Guthrie, 2017a). The consequences include significant research pressures that constrain the freedom of academics to conduct research of their choice and the need to reconcile their research interests with the strategies of their universities (Martin-Sardesai, Irvine, Tooley, & Guthrie, 2017b).

Such research measurement approaches also have induced some academics to prioritise research compared to other work. With scholarly productivity increasingly measured and audited through quantitative indicators based on PMS data such as publications and grants, academics have become pressured to publish and win research grants while focusing on their teaching responsibilities and workloads. This has increased academic workloads and stress (Martin-Sardesai et al., 2017c). Vesty et al.'s (2018) survey of Australian and New Zealand university accounting academics has evidenced the dynamics of these impacts on academics. Vesty et al. find that emotional exhaustion is common and that many academics are suffering from a risk of burnout due to high workloads, long working hours and a lack of personal recognition, which produces significant levels of staff cynicism, fatigue and job dissatisfaction.

6.4 Teaching and student-as-customer assessment

As a result of 40 years of 'metrifying' teaching and research, these two spheres of education in the AHES have frequently become disconnected (Cherastidtham et al., 2013; McKenzie et al., 2018). While university management may promulgate the teaching-research nexus as a marketing strategy, the idea is now a myth (Paltridge, Mayson and Schapper, 2012). Education has been massified This trend has been reinforced by large class sizes, customer-responsive packaging of the educational product, and the large-scale casualisation of Australian university teaching (Shah, Lewis and Fitzgerald, 2011; Parker, 2013).

Several recent studies addressing scholarly work in academia have indicated significant changes in how individual teachers' performance is evaluated (Grossi, Dobija, & Strzelczyk, 2020; Kallio, Kallio, Grossi, & Engblom, 2021). For instance, Grossi et al., 2020 find that universities and academic workers are affected by a range of pressures, including government regulations and control of the state (state pressure), the expectations of the professional norms and collegiality of the academic community (academic pressures), and the need to comply with international standards and market mechanisms (market pressures). This line of research exists at the intersection of two broader academic traditions. One addresses NPM and PMS (e.g. Parker, 2012; Steccolini, Salitere, & Guthrie, 2020); the other studies change in higher

education organisations and the academic profession (e.g. Kallio, Kallio, & Blomberg, 2020; Martin-Sardesai et al., 2019).

The past decade has consolidated the last 40 years of performance management in the contemporary Australian university context. Performance-based funding has exacerbated the trend towards management's pursuit of prestige and marketability. Rather than prioritising quality teaching and research, universities now seek higher positions in the global league tables. The role of the student experience and course and lecturer evaluation metric as a performance management tool. This has only reinforced the deferential elevation of the student to the position of a fee-paying customer.

Furthermore, as we know, customers can, without foundation, threaten expertise, independence, quality, and fitness for purpose – in the case of education, this means educator expertise, course quality, and graduate employability (Parker, 2002; Darwin, 2015; Shah, Lewis and Fitzgerald, 2011; Shah and Nair, 2012). Universities today are, arguably, too sensitive to student evaluations. What used to be a voluntary exercise is now compulsory, Shah and Nair (2012). explains the linkage between teaching scores and annual performance reviews for academic staff.

These historical trends have driven an intensely quantified performance measurement and management approach. Driving compliance and marketability focus down from organisation-wide to individual academics has had profound implications for the autonomy of universities and their academics. It arguably carries questionable motivations, orientations and outcomes concerning course content standards, the quality and rigour of assessment, grading standards, actual student attainment levels and graduate suitability for employment (Shah and Nair, 2012; Shah and Jarzabkowski, 2013).

6.5 A commercially competitive agenda

Carnegie (2022) points out that Australian universities are beset with an accounting-based obsession with metrics and rankings-based performance. His analysis of the vision and mission statements of 37 Australian public universities reveals them as having arrived at a state of macro-micro contradiction. While their public declarations report their macro-contributions to local, regional, national and global societies, their internal micro-measurement metrics-driven management control focuses them on pursuing KPIs that will enhance their global university rankings. Carnegie (2022) argues that these micro-strategies focus on driving performance metrics for academic staff and producing outcomes that build brand profiles to increase revenue.

That naked commercial agenda is exposed in Flemings' (2022) analysis of how universities transformed after the onset of Covid-19. From reports published by several prominent consulting firms, he identifies that, since the 1980s, universities have increasingly engaged management consultants following the NPM spirit of commercialising the public sector. This involvement, he argues, has often been concealed by universities because they implement business strategies that are primarily unsuited to the unique missions of a public university. This commercial agenda has increasingly been 'sold' by consultants as a necessary response to the exigencies of the pandemic. University managers have seized the opportunity to downsize staffing, slash operating costs, curtail working conditions, casualise contract work and tighten performance targets. These are neoliberal philosophies that have been imported from the private sector. However, the recent pandemic has provided an excuse for these austerity measures to become entrenched in the university's fundamental mission. Such trends have been in ample evidence for over two decades (Parker, 2002, 2011, 2012, 2013, 2020). In substance, our public universities are now, effectively, "public" organisations in name only (see, for

example, Carnegie (2022). The institutional propensity for constantly aggressively chasing higher global university rankings (GURs) increasingly makes their explicit mission and vision statements less credible, if not approaching irrelevance, due to the "loud noise" emanating from GURs, not just in Australia.

Moreover, this commercially competitive agenda has grown over recent decades into a dominant strategic focus for Australian universities. To this end, they have proliferated their internal performance metrics, redoubled their controls, and leveraged the results as marketing tools to appeal to target stakeholders. Wide now is the spectrum of metrics for reputational claims: marketplace rankings, learning and teaching indicators, research performance, community engagement, student numbers, graduate employment, publications statistics, facilities expenditure, and more. Performance "quality" has become code for quantity, speed, growth, cost control and profits (Parker, 2013, 2020).

7. Conclusions

This study reveals a significant shift in AHES profile, focus and management since the 1980s. It shows evidence of the onset and pervasiveness of the NPM measurement and audit culture resulting from Neoliberal philosophies that have consumed government and parliamentary thinking. During this period, Australian universities have experienced massive growth in student numbers as they have been let loose by Australian governments to self-fund their operations and increasingly contribute to the national economy. As a result, the past 40 years have witnessed a significant transmogrification of university missions that, while ostensibly still proclaiming their societal role, have been refocused on their corporate financial success built on a neoliberal user pays approach to national and international student recruitment and associated revenues. Simultaneously they have been observed to placate government policy agendas by their fixation on graduate employment outcomes, alignment with ruling government policy objectives of the day, and winning recognition for their performance as contributors to the national economy.

Australian universities' embracing NPM has transformed them from a former predominantly public sector entity to becoming broadly commercialised corporates or government-owned business entities. In so doing, their leadership and governance have changed to mimic their private sector corporate counterparts, a professionalised managerial class has overtaken their senior and middle-level leadership, and performance control and audit within the organisation have become the order of the day. Accounting and accountability have emerged as key players in facilitating this new university profile, role and audit culture embedded throughout higher education teaching, research and external engagement functions. The familiar accounting techniques and processes of private sector performance measurement, management, control, reporting, and audit now drive university strategies and have arguably reconditioned their core values.

The performance management and audit cultures have not only become central to university organisational level strategic thinking and behaviour. They have permeated the individual academic teaching and research staff levels. The latter has been increasingly subject to all-pervasive and intrusive performance surveillance, metrics measurement, and individual rankings with which they appear to have been willingly compliant in pursuit of job retention and career progression.

What is noteworthy in this historical reflection has been the apparent success of governments while requiring universities to compete in the global marketplace in maintaining remote control of the AHES. This has been achieved despite the actual decline in government funding of the

AHES. The recipe has been one of proliferating attention to global and national rankings, expanding arrays of metrics applied to teaching, research and engagement activities, as well as ever more intrusive 'quality' audits of teaching and research. Just as academics have become compliant with their employers' expanding array of performance management metrics and controls, universities have similarly become compliant with governments' NPM performance metrics. Given universities' enthusiastic embracing of their now emerged commercialised corporate identity, they have accordingly adopted the proliferating metrics and rankings applied in the AHES, frequently selecting those most favourable to their claimed marketplace image, brand and status ranking. Those metrics and rankings, however, reflect claimed 'quality' control tools and processes rather than actual practical impacts and risk reinforcing self-interested commercial outcomes focus rather than a contribution to the broader society. This raises the question of whether the post-1980s emergence of AHES performance management led the sector to mirror the commercial world stereotype: Australian Higher Education now appears to be about the financial bottom line.

This transmogrification of Australian university identity and role over the past forty years has had profound impacts externally on its societal role and internally on the work and roles of its academics. They have been subject to increasingly intrusive performance measurement and control systems to which they appear to have reacted with almost universal compliance. Their internal goal displacement is reflected in their pursuit of satisfactory student teaching evaluation scores, volume and ranking of journal article publications, and research grants demanded by their employers. Performance metrics have replaced original contributions to knowledge, societal critique, and community engagement as their prime concerns.

In offering this historical case study and critique of performance management in the AHES, we pose fundamental questions for the future. Has university commercial and reputational self-interest permanently replaced its public interest role? Will the now predominant role of Australian university academics as production-line manufacturers of accountingised and measurable teaching and research products continue into the long term? Have accounting academics been complicit in creating and promoting the performance control and management systems that have gradually eroded academics' and universities' broader community service role? We contend that our historical analysis connecting past to present has revealed disturbing insights into the inner workings of the university system. What of the future? As historians, we must historiographically desist from offering predictions for the future trajectory. As Carson and Carson (1998) and Parker (1999, p.20) argued:

".....it is important to recognise that history's ability to predict is limited by the uniqueness of each event or chain of events and their surrounding circumstances which are unlikely to ever be exactly replicated at a later date. Hence history offers explanation rather than prediction, laying a basis for informing present and future decisions by way of illuminating precedents...."

Further new references to add!!!!!!!!!!!!

Paltridge, T., Mayson, S. E., & Schapper, J. M. (2012). Covering the gap: Social inclusion, international students and the role of local government. *Australian Universities' Review*, *54*(2), 29 - 39.

Scharff, C. 2015. Life as an enterprise: Ten ways through which neoliberalism is experienced on an emotional level, *British Politics and Policy at LSE*.

Marginson, S. (1997) Competition and contestability in Australian higher education, 1987–1997,

Australian Universities Review, 40(1), 5–14.

Marginson, S. (1999) Harvards' of the Antipodes? Nation building universities in a global environment, Access: Critical Perspectives on Cultural and Policy Studies in Education, 18(2), 1–20.

Marginson, S. & Considine, M. (2000) The enterprise university: governance, strategy, reinvention, (Melbourne, Cambridge University Press).

<u>Carnegie</u>, <u>G.D.</u> (2023), "Global university rankings and the metrics maze", <u>Accounting</u>, <u>Auditing & Accountability Journal</u>, Vol. 36 No. 1, pp. 430-433.

Carnegie, G. and Parker, L.D. (2021) 'Rankings: irrelevant and wrong for universities: Institutions that pursue ranking success replace their missions with KPIs', *Campus Morning Mail*, 24 November. https://campusmorningmail.com.au/news/rankings-irrelevant-and-wrong-for-universities/

Carnegie, G., Guthrie, G. and Parker, L.D. (2020), 'The prevailing market strategy of marketisation is not looking durable or necessarily desirable', *Campus Morning Mail*, 16 July. https://campusmorningmail.com.au/news/to-work-out-whats-next-for-australias-public-universities-ask-academics/

Carson, P.P. and Carson, K., (1998), "Theoretically Grounding Management History as a Relevant and Valuable Form of Knowledge", Journal of Management History, Vol.4, No.1, pp.29-42.

Cherastidtham, I., Sonnemann, J., & Norton, A. (2013). The teaching-research nexus in higher education. Melbourne, Australia: Grattan Institute.

Haynes, K. 2017. Reflexivity in Accounting Research" in Hoque, Z., Parker, L.D., Covaleski, M.A. and Haynes, K. (eds) *The Routledge Companion to Qualitative Research Methods*, Routledge, Oxon and New York, 284-97

Hopper, T. (1999). Postcard from Japan: A management accounting view. *Accounting Auditing and Accountability Journal*, 12(1): 58–68.

Lama, T., Arias, P., Mendoza, K., & Manahan, J. (2015). Student Evaluation of Teaching surveys: do students provide accurate and reliable information? E-Journal of Social & Behavioral Research in Business. 6(1), 38

McKenzie, A., Griggs, L., Snell, R., & Meyers, G. D. (2018). The myth of the teaching-research nexus. *Legal Education Review*, 28, 1-20.

Parker, LD 1999, 'Historiography for the New Millennium: Adventures in Accounting and Management', *Accounting History*, vol. 4, no. 2, pp. 11-42.

REFERENCES

Australian Vice-Chancellors' Committee (1981) Academic staff development: Report of the AVCC working party, Canberra, ACT, Australian Government Publishing Service.

Bradley, D., Noonan, P., Nugent, H., & Scales, B. (2008). Review of Australian higher education final report. Canberra, Australia: Government Printing Service.

Carnegie, G.D. (2022) Global University rankings: The macro-micro contradiction in public university management, Accounting and Management Review, No. 26, pp. 77-109.

Carnegie, G.D. and Napier, C.J. (1996). "Critical and Interpretive Histories: Insights into Accounting's Present and Future Through its Past", Accounting, Auditing and Accountability Journal, Vol. 9 No. 3, pp. 7-39

Carr, E.H. (1987). What is History? Harmondsworth, Middlesex: Penguin, 2nd ed..

Commonwealth of Australia (2000) The Australian Higher Education Quality Assurance Framework, Canberra: Department of Education Science and Training (DEST).

Darwin, S., 2016. The emergence of contesting motives for student feedback-based evaluation in Australian higher education. *Higher Education Research & Development*, 35(3), pp.419-432.

Dawkins, J. (1987) Higher Education: A policy discussion paper, Canberra, ACT: Australian Government Publishing Service.

Fleming, P. (2022), "Never Let a Good Crisis Go to Waste": How Consulting Firms Are Using COVID-19 as a Pretext to Transform Universities and Business School Education, Academy of Management Learning & Education Published Online:18 Nov 2022 https://doi.org/10.5465/amle.2022.0217

Fleming, P. (2022), Dark Academia: How Universities Die, Pluto Press, London.

Hurst. B. C.. 1981. 'The Myth of Historical Evidence'. History and Theory, Vol. 20, No. 3.

Jenkins, K. (1991). Re-Thinking History. London: Routledge.

Parker, L.D., 1997. Informing historical research in accounting and management: Traditions, philosophies, and opportunities. *Accounting Historians Journal*, 24(2), pp.111-149.

Parker, L.D., 1999. Historiography for the New Millenium: Adventures in Accounting and Management. *Accounting History*, 4, p.2.

Parker, L.D., 2002. It's been a pleasure doing business with you: a strategic analysis and critique of university change management. *Critical perspectives on accounting*, 13(5-6), pp.603-619.

Parker, L., 2011. University corporatisation: Driving redefinition. *Critical perspectives on accounting*, 22(4), pp.434-450.

Parker, L.D. 2013, 'Contemporary University Strategising: The Financial Imperative', Financial Accountability & Management. vol. 29, no. 1, pp. 1-25.

Parker, L. (2020), "Australian universities in a pandemic world: transforming a broken business model?", Journal of Accounting and Organizational Change, Vol. 16 No. 4, pp. 541-548.

Previts, G.J., Parker, L.D. and Coffman, E.N., 1990a. An accounting historiography: subject matter and methodology. *Abacus*, 26(2), pp.136-158.

- Previts, G.J., LD. Parker, and E.N. Coffman. 1990b Accounting history: definition and relevance. *Abacus* 26, no. 1, pp. 1-16.
- Shah, M. and Jarzabkowski, L., 2013. The Australian higher education quality assurance framework: From improvement-led to compliance-driven. *Perspectives: Policy and Practice in Higher Education*, 17(3), pp.96-106.
- Shah, M., Lewis, I. and Fitzgerald, R., 2011. The renewal of quality assurance in Australian higher education: the challenge of balancing academic rigour, equity and quality outcomes. *Quality in Higher Education*, 17(3), pp.265-278.
- Shah, M. and Nair, C.S., 2012. The changing nature of teaching and unit evaluations in Australian universities. *Quality Assurance in Education*, 20(3), pp.274-288.
- Shah, M., Nair, S. and Wilson, M., 2011. Quality assurance in Australian higher education: historical and future development. *Asia Pacific Education Review*, 12(3), pp.475-483.
- Tosh, J. (1991). The Pursuit of History: Aims, Methods and New Directions in the Study of Modern History. London: Longman, 2nd. ed.

References

- ARC. 2012. About ARC, Vol. 2012. Canberra, Australia http://www.arc.gov.au/>.
- ARC. 2016. Engagement and Impact Assessment Consultation Paper, Vol. 2017.
- Australia, P. o. 2021. Overseas students in Australian higher education: A quick guide: Australian Parliament House (APH).
- Australia), U. U. 2017. The facts on university funding, Vol. 2020. Canberra: Universities Australia.
- Baum, F., Dollard, M., Fisher, M., Freeman, T., & Newman, L. 2022. The corporate university and its impact on health in Australia. *Social Alternatives*, 41(1): 52-62.
- Bay, U. 2011. Unpacking neo-liberal technologies of government in Australian higher education social work departments. *Journal of Social Work*, 11(2): 222-236.
- Bessant, B. 1996. Higher Education in Australia the Unified National System. *Education Research and Perspectives*, 28(1): 110-123.
- Bexley, E., James, R., & Arkoudis, S. 2011. The Australian Academic Profession in Transition. In E. a. W. R. Department of Education (Ed.). Canberra: Commonwealth of Australia.
- Brown, W. 2015. *Undoing the demos: Neoliberalism's stealth revolution*. New York: Wendy Brown Zone Books.
- Burnes, B., & Oshwick, C. 2013. Creating a Leading Journal and Maintaining Academic Freedom. *Journal of Change Management*, 13(1): 1-8.
- Canaan, J. 2010. Analysing a 'neoliberal moment' in English higher education today. *Learning* and *Teaching*, 3(2): 55-72.
- Carnegie, G., Guthrie, J., & Martin-Sardesai, A. 2021. Public universities and impacts of COVID-19 in Australia: risk disclosures and organisational change. *Accounting*, *Auditing & Accountability Journal*, forthcoming.
- Carr, K. 2008. A New Era for Australian Research Quality Assessment, Vol. 2011.

- Coaldrake, P., & Stedman, L. 1999. Academic work in the twenty-first century. *Department of Education, Training and Youth Affairs, The Commonwealth of Australia.*
- Connell, R. 2022. Raewyn Connell on why and how universities need to change, and soon: Interview with Richard Hil. *Social Alternatives*, 41(1): 76-80.
- Davies, B., Gottsche, M., & Bansel, P. 2006. The rise and fall of the neo-liberal university. *European Journal of Education*, 41(2): 305-319.
- Dawkins, J. S. 1987. Higher Education a Policy Discussion Paper. Canberra: ACT: AGPS.
- Dawkins, J. S. 1988. Higher Education a Policy Statement. Canberra, ACT: AGPS.
- DFAT. 2019. Trade and Investment at a Glance 2019. In D. o. F. A. a. Trade (Ed.). Online: Australian Government Publishing Service.
- DFT (Department of Finance). 1994. *Commonwealth Financial Management Handbook*. Canberra: AGPS.
- Ferlie, E., Ashburner, L., & Pettigrew, A. 1996. The New Public Management in Action. *Oxford University Press, New York*.
- Finkin, M., & Post, R. 2009. For the Common Good: Principles of American Academic Freedom. Michigan, USA: Yale University Press.
- Gallagher, M. 2003. The Evolution of Higher Education Financing in Australia, *German-Australian Conference on Higher Education Financing*. Berlin, 24 October, 2003.
- Gamage, D. T. 1992. Recent Reform in Australian Higher Education with Particular Reference to Institutional Amalgamations. *Higher Education*, 24: 77-91.
- Grossi, G., Dobija, D., & Strzelczyk, W. 2020. The impact of competing institutional pressures and logics on the use of performance measurement in hybrid universities. *Public Performance & Management Review*, 43(4): 818-844.
- Guthrie, J., Andrews, J., & Baker, M. 2020. 'A thimble and pea trick' submission to the Education and Employment Legislation Committee, Higher Education Support Amendment (Job-Ready Graduates and Supporting Regional and Remote Students). *Bill 2020, Second Reading*.
- Guthrie, J., & Dumay, J. 2020. Education is the answer, not the problem. *Journal of Behvaoural Economics and Social Systems*, 2(1): 7-12.
- Guthrie, J., & English, L. 1997. Performance Information and Programme Evaluation in the Australian Public System. *International Journal of Public System Management*, 10(3): 154-164.
- Guthrie, J., Evans, E., & Burritt, R. 2014. Australian accounting academics: Challenges and possibilities. *Mediatri Accountancy Research*, 22(1): 20-37.
- Guthrie, J., Evans, E., & Burritt, R. (Eds.). 2017. *Improving collaboration and innovation between industry and business schools in Australia*. (Vol. 8). Sydney: Chartered Accountants Australia and New Zealand.
- Guthrie, J., Humphrey, C., & Olson, O. 1998. International experiences with new public financial management reforms: new world? small world? better world?, *Global Warning: Debating International Developments in New Public Financial Management*: 17-48: Cappelen Adademisk Forlag.

- Guthrie, J., Linnenluecke, M., Martin-Sardesai, A., Shen, Y., & Smith, T. 2021. On the resilience of Australian public universities: Why our institutions may fail unless Vice-Chancellors rethink broken business models. *Accounting & Finance*, Forthcoming.
- Guthrie, J., & Lucas, A. 2022. How we got here: the transformation of Australian public universities into for-profit corporations. *Social Alternatives*, 41(1): 26-34.
- Guthrie, J., & Neumann, R. 2007. Economic and non-financial performance indicators in universities. *Public Management Review*, 9(1): 231-252.
- Guthrie, J., Olson, O., & Humphrey, C. 1997. Public financial management changes in OECD nations. *Advances in International Comparative Management*, 3: 225-269.
- Guthrie, J., Olson, O., & Humphrey, C. 1999. Debating developments in new public financial management: the limits of global theorising and some new ways forward. *Financial Accountability & Management*, 15(3-4): 209-228.
- Guthrie, J., & Parker, L. 1990. Public System Accounting and the Challenge of Managerialism. In J. Guthrie & L. Parker & D. Shand (Eds.)., The Public System: Contemporary Readings in Accounting and Auditing (pp. 454-469), Marrickville, NSW: Harcourt Brace Javonovich Group Australia Pty Ltd.
- Guthrie, J., Parker, L., & Shand, D. (Eds.). 1990. *The Public System: Contemporary Reading in Accounting and Auditing*. Sydney: Harcourt Brace Jovanovich.
- Halffman, W., & Radder, H. 2015. The academic manifesto: From an occupied to a public university. *Minerva*, 53(1): 165-187.
- Harman, G. 2009. *National Assessments of Research Escellance and Impact: UK and Australian Experience*. Rotterdam, The Netherlands: Sense Publisher.
- Harman, G., & Meek, V. L. 1988. Australian higher education reconstructed? analysis of the proposals and assumptions of the Dawkins green paper Armidale, NSW: University of New England.
- Harvey, D. (Ed.). 2005. A brief history of neoliberalism. Oxford: Oxford University Press.
- HEC. 1988. The Characteristics and Performance of Higher Education Canberra: AGPS.
- Henkel, M. 2005. Acdemic Identiy and Autonomy in a Changing Policy Environment. *Higher Education*, 49(1/2): 155-176.
- Hil, R., Pelizzon, A., & Baum, F. 2022. Its time: The re-form of Australian Public Universities. *Social Alternatives*, 41(1).
- Hoare, D. 1995. Review Summary of Committee Report and Recommendations, *Higher Education Management Review*. Canberra: Australian Government Publishing Service.
- Hurst, D. 2020. Universities warn funding overhaul will put quality of education in Australia at risk, *The Guardian*. Online: The Guardian.
- Kallio, K.-M., Kallio, T. J., Grossi, G., & Engblom, J. 2021. Institutional logic and scholars' reactions to performance measurement in universities. *Accounting, Auditing & Accountability Journal*, 34(9): 135-161.
- Kallio, T. J., Kallio, K. M., & Blomberg, A. 2020. From professional bureaucracy to competitive bureaucracy: Redefining universities' organisation principles, performance measurement criteria and reason for being. *Qualitative Research in Accounting & Management*, 17(1): 82-108.

- Lake, S., Joannes-Boyau, R., Lucas, A., McCallum, A., O'Connor, J., Pelizzon, A., Tregear, P.,
 & Oliver, V. 2022. A brief history of Australian universities. *Social Alternatives*, 41(1): 8-16.
- Latif, M. 2014. Academic freedom: Problems in conceptualization and research. *Higher Education Research & Development*, 33(2): 399-401.
- Liszka, J. 2008. Academic Freedom: A Basic Guide. Anchorage: University of Alaska.
- Mahony, D. 1992. Establishing the university as the sole Provider of Higher Education: The Australian Experience. *Studies in Higher Education*, 17(2): 219-236.
- Mahony, D. 1993. The Construction and Challenges of Australia's Post-Binary System of Higher Education. *Oxford Review of Education*, 19(4): 465-483.
- Marginson, S. (Ed.). 1993. *Education and public policy in Australia*. Cambridge: Cambridge University Press.
- Marginson, S. 1995. Markets in Education: A theoretical note. *Australian Journal of Education*, 39(3): 294-312.
- Marginson, S. 1997. Steering from a Distance: Power Relations in Australian Higher Education. *Higher Education* 34(63-80).
- Martin-Sardesai, A., & Guthrie, J. 2018. Accounting for construction of research quality in Australia's research assessment exercise. In S. Widener, M. Epstein, & F. Verbeeten (Eds.), *Performance Measurement and Management Control: The Relevance of Performance Measurement and Management Control Research*: 221-241. UK: Emerald Group Publishing Ltd.
- Martin-Sardesai, A., Guthrie, J., & Parker, L. 2020. The neoliberal reality of higher education in Australia: How accountingisation is corporatising knowledge. *Meditari Accountancy Research*.
- Martin-Sardesai, A., Guthrie, J., Tooley, S., & Chaplin, S. 2019. History of performance measurment systems in the Australian higher education system. *Accounting History*, 12(1): 40-61.
- Martin-Sardesai, A., Irvine, H., Tooley, S., & Guthrie, J. 2015. 'From the Global to the Local: The Impact of Australia's Research Evaluation Exercise on a University Control System, *European Accounting Association*, *38th Annual Congress*. 28-30 April 2015, Glasgow, UK.
- Martin-Sardesai, A., Irvine, H., Tooley, S., & Guthrie, J. 2017a. Accounting for research: academic responses to research performance demands in an Australian University. *The Australian Accounting Review*, 27(3): 329-334.
- Martin-Sardesai, A., Irvine, H., Tooley, S., & Guthrie, J. 2017b. Government research evaluations and academic freedom: a UK and Australian comparison. *Higher Education Research & Development*, 36(2): 372-385.
- Martin-Sardesai, A., Irvine, H., Tooley, S., & Guthrie, J. 2017c. Organizational change in an Australian university: Responses to a research assessment exercise. *The British Accounting Review*, 49(4): 399-412.
- Morrish, L. 2017. *Neoliberalism and the Academy: Have you drunk the Kool-Aid?* Basingstoke: Palgrave Macmillan.

- Norton, A., & Cherastidtham, I. 2018. *Mapping Australian higher education* Carlton, Victoria: Grattan Institute.
- Olssen, M., & Peters, M. A. 2005. Neoliberalism, Higher Education and the Knowledge Ecology: From the Free Market to Knowledge Capitalism. *Journal of Education Policy*, 200(3): 313-345.
- Parker, L. 2012. From Privatised to Hybrid Corporatised Higher Education: A Global Financial Management Discourse. *Financial Accountability & Management*, 28(3): 247-268.
- Parker, L., & Guthrie, J. 1993. The Australian public system in the 1990s: new accountability regimes in motion. *Journal of International Accounting, Auditing and Taxation*, 2(1): 59-81.
- Parker, L., Martin-Sardesai, A., & Guthrie, J. 2021. Four decades of neoliberalim and Australian public universities: an accountability transformation by stealth. *Financial Accountability & Management*, forthcoming.
- Parker, L. D., & Guthrie, J. 1998. Managerialism and marketisation in financial management change in Australia: Cappelen Akademisk Forlag As.
- Power, M. 1997. *The Audit Society: Rituals Of Verification*. New York: Oxford University Press.
- Ryan, S., & Guthrie, J. 2009. Collegial entrepreneurialism: Australian graduate schools of business. *Public Management Review*, 11(3): 317-344.
- Scharff, C. 2015. Life as an enterprise: Ten ways through which neoliberalism is experienced on an emotional level, *British Politics and Policy at LSE*.
- Shore, C., & Wright, S. 2004. Whose accountability? Governmentality and the auditing of universities. *Parallax*, 10(2): 100-116.
- Shore, C., & Wright, S. 2015. Governing by numbers: audit culture, rankings and the new world order. *Social Anthropology. Anthropologie Sociale*, 23(1): 22-28.
- Shore, C., Wright, S., Amit, V., Brown, J., Bruun, C., Macguire, M., Merry, S., Sauder, M., Espeland, W., Smith, G., & K.Teghtsoonian. 2015. Audit culture revisited: Rankings ratings and the reassembling of society. *Current Anthropology*, 56(3): 431-432.
- Sims, M. 2022. Academic Leadership in a Neoliberal Managerial World: An autoethnography of my career. *Social Alternatives*, 41(1): 70-75.
- Smith, R. H. T. 1989. Report of Committee to Review Higher Education Research. Canberra, ACT.
- Steccolini, I., Salitere, I., & Guthrie, J. 2020. The role(s) of accounting and performance measurement systems in contemporary public administration. *Public Administration*, 98(1): 3-13.
- Tarquinio, L., Raucci, D., & Benedetti, R. 2018. An Investigation of Global Reporting Initiative Performance Indicators in Corporate Sustainability Reports: Greek, Italian and Spanish Evidence. *Sustainability*, 10(4): 897.
- Tregear, P., Guthrie, J., Lake, S., Lucas, A., O'Connor, J., Pelizzon, A., & Vodeb, O. 2022. 'Enough to make you sick!'Pathological characteristics of the Australian academic workplaces. *Social Alternatives*, 41(1): 44-51.
- U21, U. 2022. U21 Ranking of National Higher Education Systems. In J. Dixon (Ed.), Vol. 2022. Online: The University of Melbourne.

Vodeb, O., Joannes-Boyau, R., Lake, S., Lucas, A., McCallum, A., O'Connor, J., Pelizzon, A., & Tregear, P. 2022. Australian public universities and the destruction of the academic community. *Social Alternatives*, 41(1): 35-43.